Author's response to reviews

Title: Primary parotid gland lymphoma: A case report

Authors:

Petros Konofaos (petros_konofaos@yahoo.com)
Eleftherios Spartalis (eleftherios.spartalis@gmail.com)
Paraskevas Katsaronis (pkatsaronis@yahoo.gr)
Grigoris Kouraklis (gkouraklis@hotmail.com)

Version: 4 Date: 1 February 2011

Author's response to reviews: see over
Corrections list

1. Please make sure that the Abstract has section headings.

Section headings in the Abstract were added.

2. Please include a Competing Interest section. Please declare whether competing interest exists.

A Competing interest section was added.

3. Please include a section for abbreviations.

A section for Abbreviations was added.

4. Please include an Authors' Contributions section.

An Authors' Contributions section was added.

5. Please include a Patient Consent section.

A Patient Consent section was added.

6. Please send a copy of the signed consent form to the editorial office.

A signed consent form will be sent to the Editorial office.

7. Please remove the date and time from Figure 1.

Date and time were removed from Figure 1.